

GOPEN ACCESS

Citation: Greenbaum G, Feldman MW, Rosenberg NA, Kim J (2021) Designing gene drives to limit spillover to non-target populations. PLoS Genet 17(2): e1009278. https://doi.org/10.1371/journal.pgen.1009278

Editor: Lindi Wahl, University of Western Ontario, CANADA

Received: July 27, 2020

Accepted: November 14, 2020

Published: February 25, 2021

Copyright: © 2021 Greenbaum et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> information files. The code for generating the numerical solutions described in this paper can be downloaded as a Mathematica notebook from github.com/GiliG/Gene_Drive_DTE.

Funding: This study was supported by NIH grant R01HG005855 awarded to NAR and by the Center for Computational, Evolutionary and Human Genomics (CEHG) at Stanford. The funders had no role in study design, data collection and analysis, **RESEARCH ARTICLE**

Designing gene drives to limit spillover to nontarget populations

Gili Greenbaum^{1*}, Marcus W. Feldman², Noah A. Rosenberg², Jaehee Kim²

1 Department of Ecology, Evolution, and Behavior, The Hebrew University of Jerusalem, Jerusalem, Israel, 2 Department of Biology, Stanford University, Stanford, California, United States of America

* gil.g@mail.huji.ac.il

Abstract

The prospect of utilizing CRISPR-based gene-drive technology for controlling populations has generated much excitement. However, the potential for spillovers of gene-drive alleles from the target population to non-target populations has raised concerns. Here, using mathematical models, we investigate the possibility of limiting spillovers to non-target populations by designing differential-targeting gene drives, in which the expected equilibrium gene-drive allele frequencies are high in the target population but low in the non-target population. We find that achieving differential targeting is possible with certain configurations of gene-drive parameters, but, in most cases, only under relatively low migration rates between populations. Under high migration, differential targeting is possible only in a narrow region of the parameter space. Because fixation of the gene drive in the non-target population could severely disrupt ecosystems, we outline possible ways to avoid this outcome. We apply our model to two potential applications of gene drives—field trials for malaria-vector gene drives and control of invasive species on islands. We discuss theoretical predictions of key requirements for differential targeting and their practical implications.

Author summary

CRISPR-based gene drive is an emerging genetic engineering technology that enables engineered genetic variants, which are usually designed to be harmful to the organism carrying them, to be spread rapidly in populations. Although this technology is promising for controlling disease vectors and invasive species, there is a considerable risk that a gene drive could unintentionally spillover from the target population, where it was deployed, to non-target populations. We develop mathematical models of gene-drive dynamics that incorporate migration between target and non-target populations to investigate the possibility of effectively applying a gene drive in the target population while limiting its spillover to non-target populations ('differential targeting'). We observe that the feasibility of differential targeting depends on the gene-drive design specification, as well as on the migration rates between the populations. Even when differential targeting is possible, as migration increases, the possibility for differential targeting disappears. We find that differential targeting can be effective for low migration rates, and that it is sensitive to the design of the gene drive under high migration rates. We suggest that differential targeting decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

could be used, in combination with other mitigation measures, as an additional safeguard to limit gene drive spillovers.

Introduction

Gene drives are genetic constructs that can bias transmission of desired alleles to progeny, allowing these alleles to rapidly increase in frequency even when they are negatively selected. Gene drives, therefore, have the potential to modify or even eradicate entire species. The population genetics of gene drives, under various genetic architectures, have been studied for several decades [1–9]. With recent innovation in technical engineering of gene drives using CRISPR/Cas9-based methods [10], gene drives have attracted considerable attention for their potential applications. In particular, engineered gene drives can conceivably alter or suppress disease vectors, agricultural pests, or invasive species [11–15].

However, the potential of this technology also raises significant concerns due to the possibility of gene-drive spillovers to non-targeted populations [11, 16–19]. The effects of such spillovers could be devastating, unintentionally driving species to extinction or permanently modifying important traits, potentially leading to ecological cascades [17]. With invasive-species control, there is particular concern, because every invasive species is non-invasive in its native range. Moreover, because an invasion has occurred, it is likely that invaded regions are connected to native ones through migration. As a result, to prevent gene-drive spillovers, every application of a gene drive in an invaded region must be designed to avoid them. For example, it has been suggested that CRISPR-based gene drives could be applied in New Zealand to eradicate invasive species, such as Australian possums, stoats, and rats [15, 17, 19–21]. However, such plans must account for the possibility and potential consequences of spillovers of gene drives from New Zealand to the native ranges of these species. Therefore, understanding the dynamics of gene drives with CRISPR-based constructs under migration, in the context of spillovers to non-target populations, is crucial.

Currently, due to the risks and complications of deploying gene drives, studying spillovers relies on mathematical and computational modeling. CRISPR-based gene-drive models belong to a broader class of preferential-transmission models—systems in which Mendel's law of equal segregation is violated, with preferential transmission of particular alleles to subsequent generations [22]. Because many general principles of population-genetic theory are violated by non-Mendelian segregation [6, 23], various types of preferential-transmission models have been studied extensively [1–3, 5, 24–28]. Of particular interest are cases in which alleles with enhanced transmission cause reduced fitness of individuals that carry them. Analyses of such models have focused on various types of genetic architectures and evolutionary models [12], such as meiotic drive, modifier genes, and sex-ratio distorters [3, 5, 27, 28].

CRISPR-based gene drives are particular cases of preferential transmission of alleles, and are therefore modeled with the specific features of the CRISPR mechanism. In CRISPR scenarios, the preferential transmission is generated at the zygote formation stage or in the germline by conversion of heterozygotes carrying one copy of the gene-drive allele to homozygotes with two copies of the gene-drive allele [9, 10]. This conversion occurs when the CRISPR mechanism, which is incorporated in the gene-drive allele, edits the other chromosome to replace the wild-type allele with the gene-drive allele.

One of the features of preferential-transmission models is the existence of polymorphic equilibria, namely the states where preferred alleles persist in the population together with wild-type alleles, rather than sweeping to fixation or loss. These equilibria can be either stable

or unstable, depending on the genetic architecture involved [1, 2, 4-6, 27, 29]. With CRISPRbased gene-drive technology, it has been suggested that, in a single isolated population, unstable polymorphic equilibria could be utilized for generating a biosafety measure to address the dangers of accidental releases [9]. If a gene drive is initiated at a frequency below an unstable equilibrium, it is expected to be driven to loss, whereas it is expected to be driven towards fixation if initiated at frequencies above the equilibrium [1, 9, 30-32]. Similarly, in spatially continuous populations, it has been argued that a CRISPR-based gene drive with unstable equilibria can be engineered to be driven to fixation only once it is introduced over a sufficiently large area [33].

However, few natural populations exist in isolation, and distinct populations are often connected via migration. Therefore, in order to understand potential consequences of gene-drive spillovers, explicit incorporation of migration between populations into gene-drive models is required. Some studies have examined preferential-transmission migration models with genetic architectures that have not been CRISPR-based [8, 34–39]. Among these studies, some have found that polymorphic equilibria can exist for low migration rates, but not necessarily for high migration rates.

Polymorphic equilibrium states that represent stable conditions under which gene-drive allele frequencies are high in one population but low in another might potentially be exploited to mitigate spillovers. This approach would require (1) identifying stable states in which gene-drive frequencies are high in the target population and low in the non-target population, (2) configuring the genetic architecture of a CRISPR-based gene drive to attain these states, and (3) initiating the gene drive such that it would converge to these states. We term this approach *differential targeting*. In order to consider the prospect of mitigating gene-drive spillovers through differential targeting, gene-drive models that incorporate migration and CRISPR-based genetic architectures are required.

Here, we develop and investigate such models in the context of gene-drive spillovers. We focus on identifying gene-drive designs that allow for differential targeting, and evaluate the feasibility of the approach for mitigating spillovers.

Results

Modeling CRISPR-based gene drives with migration

The dynamics of gene drives introduced into a wild population depend in part on features that can, at least in principle, be configured by researchers, such as the gene-drive phenotype and the conversion rate, and in part on features that cannot be controlled, such as the ecological circumstances and life-history traits of the species. In particular, the migration levels between populations are not a controlled feature of the gene drive (unless the gene drive targets dispersal-related traits). Here, in addition to migration, we consider several features of gene drives: (1) the selection coefficients of individuals carrying the gene-drive allele, coefficients that are related to the designed gene-drive phenotype; (2) the life-stage at which the gene-drive phenotype is expressed and subjected to natural selection, specifically whether selection acts before or after the typical migratory life stage or the gene drive conversion; (3) the degree of dominance of the gene-drive allele relative to the wild-type allele; (4) the efficiency of the gene-drive conversion mechanism.

We model a CRISPR-based gene drive [10], following previously described models [7, 9]. We consider a population with a wild-type allele, *a*, and a gene-drive allele, *A*, which is initially absent from the population. The gene drive is characterized by (1) the conversion rate *c* of heterozygotes carrying the gene-drive allele to homozygotes of the gene-drive allele (from *Aa* to *AA*), with c = 1 being full conversion and c = 0 being regular Mendelian inheritance; (2) the degree of dominance

h of the gene-drive allele; (3) the selection coefficient *s* of a homozygote for the gene-drive allele relative to the wild-type homozygote. In other words, the fitnesses are 1 - s for the *AA* genotype, 1 - hs for the *Aa* genotype, and 1 for the *aa* genotype. For s > 0, the gene-drive allele is negatively selected, and we discuss only this case here, as a beneficial gene drive (s < 0) is expected to be driven to fixation in a deterministic model in all connected subpopulations. The evolutionary dynamics of a gene drive can be formulated as a recursion equation describing the change in the frequency of the gene-drive allele, *q*, over one generation [7, 9] (Fig 1A):

$$q' = \frac{q^2(1-s) + 2q(1-q)(s_n + s_c)}{\bar{w}}.$$
(1)

Here, s_n is the contribution to selection of non-converted heterozygotes, s_c is the contribution of converted heterozygotes, and \bar{w} is the mean fitness of the population (see Appendix A in <u>S1 Text</u> for a detailed description of the one-deme model). In this model, $s_n = \frac{1}{2}(1-c)(1-hs)$, $s_c = c(1-s)$ if conversion occurs prior to selection (e.g., conversion in the zygote [10]), and $s_c = c(1-hs)$ if conversion occurs after selection [7] (e.g., in the germline [40]); we consider here the case of conversion occurring before selection, and we address conversion after selection in Appendix C in <u>S1 Text</u>.

Two-deme model. To incorporate migration between the target population and a nontarget population into this modeling framework, we extend the one-deme model in Eq 1. We assume two connected demes, each large and panmictic, with symmetric migration at a rate mbetween them (Fig 1B). As in the one-deme model, the dynamics of the gene drive follow the changes in allele frequencies of the A allele in the two demes, q_1 and q_2 , respectively. Here, we study a model in which migration occurs before selection, meaning that individuals migrate at a relatively early life stage, and the fitness consequences of the gene-drive phenotypes are expressed in the deme to which the individuals have migrated (e.g., phenotypes expressed at late life stages, such as during reproduction). We have also analyzed two alternative models: selection occurring before migration (Appendix B in S1 Text and S1 Fig), and conversion occurring after migration and selection (Appendix C in S1 Text).





https://doi.org/10.1371/journal.pgen.1009278.g001

We first consider the allele frequencies of *A* after migration but before selection, \tilde{q}_1 and \tilde{q}_2 in demes 1 and 2, respectively. These frequencies are obtained by accounting for the relative contributions to the *A* allele frequencies of residents and migrants (Fig 1C, black arrows):

$$\tilde{q}_1 = (1 - m)q_1 + mq_2$$

$$\tilde{q}_2 = (1 - m)q_2 + mq_1.$$
(2)

Next, we consider the frequencies of *A* after selection takes place on the post-migration gene pools (Fig 1C, orange arrows). The changes in the allele frequencies between generations in each population are obtained as in Eq 1, using the post-migration frequencies from Eq 2:

$$q_{1}' = \frac{\tilde{q}_{1}^{2}(1-s) + 2\tilde{q}_{1}(1-\tilde{q}_{1})(s_{n}+s_{c})}{\bar{w}_{1}}$$

$$q_{2}' = \frac{\tilde{q}_{2}^{2}(1-s) + 2\tilde{q}_{2}(1-\tilde{q}_{2})(s_{n}+s_{c})}{\bar{w}_{2}},$$
(3)

where \bar{w}_1 and \bar{w}_2 are the post-migration mean fitnesses of demes 1 and 2, respectively, expressed as $\bar{w}_i = \tilde{q}_i^2(1-s) + 2\tilde{q}_i(1-\tilde{q}_i)(2s_n+s_c) + (1-\tilde{q}_i)^2$.

Differential-targeting equilibria (DTEs)

In order to understand the evolutionary trajectories of the gene drive in the two-deme system, we study the equilibrium states of the model. This is accomplished by solving Eq 3 under the equilibrium conditions $q'_1 = q_1$ and $q'_2 = q_2$. We denote these solutions, the equilibrium points in frequency space, by (\hat{q}_1, \hat{q}_2) , where \hat{q}_1 and \hat{q}_2 are the equilibrium frequencies in demes 1 and 2, respectively. We consider only solutions for which both \hat{q}_1 and \hat{q}_2 lie in the interval [0, 1], and we analyze the stability of these equilibria (see Methods).

A gene-drive configuration is denoted (*s*, *c*, *h*), where *s*, *c*, and *h* can have any value in the interval [0, 1]. Under any gene-drive configuration, there are two trivial equilibrium points, corresponding to global fixation, $(\hat{q}_1, \hat{q}_2) = (1, 1)$, and global loss, $(\hat{q}_1, \hat{q}_2) = (0, 0)$, of the gene-drive allele. In the one-deme model, it has been demonstrated that for some gene-drive configurations (*s*, *c*, *h*), there exists a single non-trivial (polymorphic) equilibrium point, which could be either stable or unstable [7, 9]. For these configurations, the stabilities of the equilibria are alternating, i.e., the two trivial equilibria are stable and the non-trivial equilibrium is stable.

In order to understand how different gene-drive configurations result in different equilibrium states, we partition the set of possible configurations of (s, c, h) into four subsets (S3–S9 Figs) based on their behaviors in the one-deme model (Eq 1). The first two subsets consist of those configurations for which there exist only the two trivial equilibria, one of which is stable: (1) A_1 , the set of configurations for which the gene-drive allele eventually reaches fixation, and (2) A_2 , the set of configurations for which the gene-drive allele is eventually lost. The other two subsets are those for which there is a non-trivial equilibrium, in addition to the two trivial ones: (3) B_1 , the set of configurations for which the additional non-trivial equilibrium is stable, and (4) B_2 , the set of configurations for which the additional non-trivial equilibrium is unstable. The B_2 configurations are of particular interest, because they represent threshold-dependent gene drives, which spread only if initiated at frequency above the unstable equilibrium [7, 9].





We can leverage the results obtained for the one-deme model to explore equilibria in the two-deme model. For the two-deme model, the number of equilibrium solutions for Eq 3 depends on the gene-drive configurations (*s*, *c*, *h*), and also on the migration rate *m* (Fig 2). For tractability, we first consider an ecologically uninteresting yet illustrative case, where m = 0. In this case of no migration, the two-deme system becomes two independent one-deme systems. For B_1 or B_2 configurations, the number of equilibria in each one-deme system is 3, and therefore, the number of equilibria in the two disconnected demes is $3 \times 3 = 9$ (for example, Fig 2A). Of these equilibria, one equilibrium point, (1, 0), represents a desired outcome, in which the target deme (population 1) is affected by the gene drive, which is absent from the non-target deme (population 2). This equilibrium is stable only for B_2 configurations, in each of the one-deme systems and hence in the two-deme system, and it is unstable for B_1 configurations. Therefore, if a gene drive with a configuration in B_2 is initiated in the basin of attraction of the stable equilibrium point (1, 0) (e.g., yellow region in Fig 2A), then we expect that the desired outcome—the gene drive sweeping the target population and not the non-target population—will be reached.

Our main interest is to understand whether similar desired outcomes, in which a stable equilibrium exists with high gene-drive frequencies in the target population and low frequencies in the non-target population, can exist for m > 0. We term such an equilibrium point (\hat{q}_1, \hat{q}_2) , for which $\hat{q}_1 > \hat{q}_2$, a *differential-targeting equilibrium* (DTE). In other words, we define an equilibrium point as a DTE if it is stable, and if the equilibrium gene-drive frequency in the target deme is larger than the frequency in the non-target deme. If a DTE exists, and we initiate the gene drive in its basin of attraction, then the system converges to frequencies that are higher in the target population than in the non-target population.

For m > 0, we obtain the possible number and type of equilibria attainable for Eq.3 by numerically exploring the parameter space of possible gene-drive configurations and

migration rates. In previous work for the specific case of m = 0, it has been seen that if (1, 0) or (0, 1) are stable equilibria, then stable non-trivial equilibria will exist if m > 0 is small enough [34, 35]. Although the gene-drive configuration sets A_1, A_2, B_1 , and B_2 were defined for the one-deme model rather than for the two-deme model, and they are independent of m, they play an important role for investigating the existence of DTEs. For B_2 configurations, and only for B_2 configurations, we numerically find that 9 equilibria exist for some m > 0. Moreover, 9 equilibria exist only for low m > 0, whereas for higher migration rates there exist fewer equilibria (S4–S9 Figs). For example, 9 equilibria exist in Fig 2A–2F, but only 5 exist in Fig 2G–2J. Only when 9 equilibria exist does there exist a DTE, and this DTE is always unique. For example, in Fig 2A–2F, the yellow points are DTEs, and the yellow regions around them are the corresponding basins of attraction. At the DTE, the gene drive is maintained at a considerably lower frequency in the non-target population than in the target population.

Notably, we observe that DTEs exist only for low migration rates (S4–S9 Figs). For B_2 configurations, when migration exceeds a certain critical threshold, there are no longer 9 equilibrium solutions to Eq.3, but only 5 or 3 solutions, none of which are both stable and allow differential targeting (S4–S9 Figs). We label this critical threshold of existence of a DTE by $m^*(s, c, h)$ (Fig 2). In other words, $m^*(s, c, h)$ is defined as the supremum of the set of migration rates m for which there are 9 equilibria for Eq.3 with the parameters s, c, h, and m. This set of migration rates is not empty, because for m = 0 there are 9 equilibria for B_2 configurations, as shown above, and therefore the supremum m^* is well-defined. For $m > m^*$, we observe that all stable equilibria are symmetric ($\hat{q}_1 = \hat{q}_2$). For (s, c, h) configurations that are not in B_2 , m^* is undefined, because in these cases no DTEs exist for any m > 0.

We investigated m^* numerically across the parameter space. We observe that for scenarios where $m > m^*$, only two stable equilibria exist—global fixation (1, 1) and global loss (0, 0). Hence, differential-targeting of a gene drive is not possible if $m > m^*$, and spillover to deme 2 of a gene drive that affects deme 1 is unavoidable. For A_1, A_2 , and B_1 configurations and for any m > 0, the results in the two-deme models are equivalent to those in the one-deme model. In other words, for each equilibrium \hat{q}_i in the one-deme model, (\hat{q}_i, \hat{q}_i) is the corresponding equilibrium in the two-deme model, with the same stability properties.

Migration and differential targeting

We obtained m^* numerically for all configurations (s, c, h) in B_2 using the numerical equilibrium solution for Eq 3 (Fig 3A–3C). We find that for most of the parameter range, m^* is low (blue regions in Fig 3A–3C), except in a narrow curved band across the parameter space (pale yellow bands in Fig 3A–3C). m^* increases for higher conversion rates c, and is maximal ($m^* \approx 0.110$) for c = 1 and $s \approx 0.72$; we denote this maximizing selection coefficient by s^* . Note that h values are not relevant for a full-conversion gene drive (c = 1), because in such configurations, the system has no heterozygous individuals.

Impact of differential targeting on the non-target population. In practice, as the migration rate *m* cannot be estimated accurately and may vary over time, it is important to understand the potential consequences for the non-target population of differential targeting, for a range of values of *m*. In principle, assuming the goal is suppression of the population in deme 1, initiation of the gene drive in the basin of attraction of the DTE leads to a high frequency of *A* in deme 1 and a low frequency in deme 2. This state is maintained until the population in deme 1 begins to collapse due to the population-level impact of the gene drive. For these generations, the population in deme 2, the non-target population, experiences the burden of the gene drive at frequency \hat{q}_2 of the DTE.



Fig 3. Critical migration thresholds for differential-targeting in the two-deme model. (A–C) Maximal migration rates m^* for which a DTE exists. The colored regions denote the configurations for which a DTE exists (B_2 configurations), and the regions in white denote configurations for which differential targeting of the demes is not possible (A_1 , A_2 , and B_1). Over most of the parameter space, a DTE exists only for low migration rates (in blue), and only in a narrow band do DTEs exist with migration rates above $m^* = 0.05$ (light yellow). (D–F) Maximal gene-drive frequencies in the non-target population, q_2^* , at DTEs. q_2^* values are correlated with m^* values.

For a given configuration (s, c, h) in B_2 , we define the supremum equilibrium frequency of the DTE in the non-target population for migration rates below m^* as $q_2^*(s, c, h) = \sup_{m < m^*} \hat{q}_2(s, c, h, m)$, where $\hat{q}_2(s, c, h, m)$ is the gene-drive frequency at the DTE with configuration (s, c, h) under migration rate m. We computed q_2^* numerically across the parameter range B_2 (Fig 3D–3F). In general, we find that q_2^* is positively correlated with m^* (Fig 3), meaning that gene-drive configurations that can sustain differential targeting for higher migration rates also potentially result in higher frequencies of the gene drives in the non-target population. Considering all possible gene-drive configurations in B_2 , q_2^* is maximal for c = 1 and s = s^* , the same configuration that maximizes m^* . This maximal allele frequency of A in the nontarget population at a DTE is $q_2^* = 0.28$. The configurations that produce these q_2^* values, with c = 1 and $s = s^*$, delineate the conditions at which differential targeting influences the non-target population most strongly in terms of frequency of the gene-drive allele. Notably, if c = 1, then q_2^* falls sharply for $s > s^*$ and less so for $s < s^*$ (Fig 3D–3F), suggesting that deviation from s^* is not symmetric in its impact on deme 2.

Similar results in terms of existence of DTEs, m^* , and q_2^* were obtained for the alternative model with the order of selection and migration reversed (Appendix B in S1 Text and S2 Fig).

The values for the alternative model in Eq S2 only slightly differed from those for the model in Eq 3. In the models where conversion occurs after selection (Appendix C in S1 Text), we observe that DTEs are not attainable for any gene drive with h = 0 and h = 0.5, and that m^* and q_2^* are identical to those in models with conversion before selection for h = 1.

Exceeding the critical migration rate m^* . For a given (*s*, *c*, *h*) configuration, the limiting migration rate at which the possibility for differential targeting can withstand migration is m^* . However, we must also consider the consequences of applying gene drives when the migration rate exceeds m^* , because the migration rate may unexpectedly change over time and may be difficult to measure. For a given configuration, if $m > m^*$, then the two-deme system is expected to rapidly converge either to global loss or to global fixation of the gene-drive allele. A gene-drive configuration that converges to global loss if m^* is exceeded results in a failure of the gene-drive application. This case would, in general, be preferable to a gene drive that converges to global fixation when migration exceeds m^* , as the latter case would likely result in severe consequences for the non-target population.

To explore the outcomes for exceeding the threshold m^* , we assume that when m changes, the system converges rapidly to its equilibrium state, and we ignore the transient dynamics of changes between stable states. With this assumption, we can consider the continuous changes in the stable equilibrium point in frequency space due to continuous changes in the migration rate. We track the trajectory of the DTE by varying m from 0 to m^* for each gene-drive configuration (s, c, h). To investigate the consequence of a breach of m^* , we continue this trajectory above m^* with an instantaneous transition to either global fixation or global loss. The state to which the trajectory transitions was evaluated by determining whether (\hat{q}_1, \hat{q}_2), computed for the DTE with $m = m^* - \epsilon$, lies in the basin of attraction of global loss or of global fixation for the DTE computed for $m = m^* + \epsilon$. For example, in Fig 2, the yellow DTE for m = 0.022, just below m^* , is positioned in the red global-fixation basin of attraction for the scenario m = 0.023, just above m^* . We therefore conclude that, for the configuration in Fig 2, the gene drive would converge to global fixation if m^* were exceeded.

We computed trajectories of DTEs with an increase in migration rates for c = 1 and different selection coefficients s (Fig 4). We observe that for selection coefficients s lower than s^* , exceeding m^* results in global fixation, whereas for selection coefficients higher than s^* , breaching the m^* threshold results in global loss. This sharp transition, and the severity of the consequences of global fixation, suggest that a safer gene-drive configuration is one with a somewhat higher selection coefficient than s^* ; maintaining a safe distance from this sharp transition would be prudent, although at a cost of producing a decrease in m^* . The qualitative behavior of the system with respect to breaching m^* is similar for c < 1, with a different threshold selection coefficient s that distinguishes global fixation from global loss.

Perturbations from the DTEs. So far, we have evaluated gene-drive spillovers by considering the equilibria to which the dynamics converge deterministically. However, we have not considered the possibility that stochastic events and processes might destabilize the system, allowing it to escape the DTE due to perturbations, and to reach a different stable state. In this section, we consider two types of perturbations: (1) perturbations due to genetic drift, and (2) perturbations due to an external event, such as a significant ecological disturbance, which can affect the allele frequencies in the demes.

The impact of genetic drift on the probability of escaping a DTE depends on the effective population size of each deme, N_e . We estimated the *probability of escape*—the probability of leaving the basin of attraction of the DTE in 100 generations, starting at the DTE—by simulating the dynamics of models similar to Eq 3, but with allele frequencies also affected by genetic drift (see Appendix D in <u>S1 Text</u> for details). We examined the scenario of $N_e = 100$ and c = 1 as an example (recall that *h* has no significance for c = 1 because there are no heterozygotes).



Fig 4. Trajectories of DTEs with increasing migration rates. Shown are trajectories for DTEs for gene drives with full conversion (c = 1; h values are not relevant in this case since there are no heterozygotes) and different selection coefficients (s), marked in different colors. Arrows show the direction of increased migration; solid lines show the increase of migration below m^* , and dashed lines show the abrupt transition of the equilibrium state when m^* is exceeded, to either global fixation (1, 1) or global loss (0, 0). The color bars for s are centered such that the selection coefficient that maximizes m^* , $s^* \approx 0.72$, is shown in black, selection coefficients below s^* are in red, and those above s^* are in blue. s^* forms the threshold between convergence to global fixation and convergence to global loss when the threshold m^* is exceeded.

In general, for this scenario, the probability of escape remains below 5% until m^* is approached (Fig 5A and S10(A) Fig). For larger populations, with $N_e > 100$, the probabilities of escape are expected to be lower than those in Fig 5A and S10(A) Fig due to weaker genetic drift, whereas for populations with $N_e < 100$, the probabilities of escape are higher (S11 Fig).

Genetic drift is a source of perturbation that is intrinsic to the system, but we also consider extrinsic perturbations that may alter allele frequencies in the demes. In this case, we evaluate the resilience to perturbations by examining the effect of a single perturbation on the eventual DTE of the system. We define the safety radius of a gene-drive configuration under a given migration rate to be the maximal magnitude in the allele frequency space of a perturbation from a DTE above which the system will not converge back to the DTE, i.e., the shortest Euclidean distance from the DTE to the boundary of its basin of attraction (details in Appendix D in S1 Text). For perturbations from the DTE larger than the safety radius, the system is expected to converge to an equilibrium that is not a DTE. Similar to the probability of escape from the DTE by genetic drift, the safety radii are particularly small when migration rates approach m^* (Fig 5B and S10(B) Fig).

Asymmetric migration. The analysis above has been conducted under the assumption that migration between the two populations is symmetric. If outgoing migration is assumed to be proportional to the population size or density, then the symmetry assumption could be interpreted as an implicit assumption of equal population sizes. In this section, we explore the



Fig 5. Perturbation from DTEs. Results shown for gene drives with full conversion, c = 1, and for different migration rates *m* and selection coefficients *s*. White regions denote scenarios for which $m > m^*$. (A) The probability of escape from the DTE due to genetic drift, defined as the probability of departing from the attraction basin of the DTE over 100 generations with genetic drift, in a Wright-Fisher population with $N_e = 100$. The black line denotes 5% probability of escape. Probabilities were estimated with 1000 simulated replicates. (B) The safety radius of the DTE.

effect of asymmetry in migration by supposing that the parameter *m* represents the migration rate from deme 2, the non-target population, to deme 1, the target population, and that migration from deme 1 to deme 2 occurs at rate *am*. Therefore, *a* is the ratio between the migration rates in the two directions. Analogously to Eq 2, the post-migration gene drive frequencies are given by:

$$\tilde{q}_{1} = [(1 - am)q_{1} + mq_{2}]\frac{1}{1 - am + m}$$

$$\tilde{q}_{2} = [(1 - m)q_{2} + amq_{1}]\frac{1}{1 - m + am},$$
(4)

where in each equation, the frequencies are normalized to account for the differences in the two migration rates under the assumption that population sizes are density-regulated and are fixed. To avoid nonsensical migration rates, we only address scenarios for which am < 1. We obtain the equilibrium conditions for this asymmetric two-deme model by solving Eq.3, using Eq.4 instead of Eq.2 for the post-migration frequencies. With a = 1, the asymmetric model reduces to the symmetric model (Eq.2).

Analogously to the symmetric model, we define $m_a^*(s, c, h)$ as the critical migration rate for existence of DTEs in the model with asymmetric migration ratio *a*. Equivalently, we define $q_{2,a}^*(s, c, h)$ as the maximal DTE frequency in the non-target population with asymmetric migration ratio *a*. Note that m_a^* describes the critical migration rate in terms of migration from deme 2 to 1; at the critical threshold, migration from deme 1 to 2 is, therefore, am_a^* .

We numerically computed m_a^* and $q_{2,a}^*$ for different migration ratios, a = 10, 2, 0.5 and 0.1, and for different gene drive configurations (Fig 6 for additive gene drives with h = 0.5; results for h = 0 and h = 1 are presented in Appendix E in S1 Text and S12 and S13 Figs). For a = 10



Fig 6. Critical migration thresholds for differential targeting, m_a^* , and maximal gene drive frequencies in the non-target population, $q_{2,a}^*$, with asymmetric migration, for additive gene drives (h = 0.5). (A–E) Critical migration thresholds from the non-target deme to the target deme, m_a^* , for migration ratio a. (F–I) Critical migration thresholds expressed as migration from target to non-target demes, am_a^* , for migration ratio a. (J–N) Maximal gene-drive frequencies in the non-target population at the at DTE, $q_{2,a}^*$, with migration ratio a.

and a = 2, we observe that $am_a^* < 1$ for all gene drive configurations we examined, ensuring that the scenarios we analyze are valid.

When migration rates are lower from the non-target population to the target population (a > 1), the critical thresholds m_a^* are substantially lower than in the symmetric case (Fig 6A–6B compared to Fig 6C). However, in terms of migration rates from the target to non-target population, the thresholds am_a^* are fairly similar to the thresholds in the symmetric case (Fig 6F–6G compared to Fig 6C), although slightly lower; the maximal critical threshold across the parameter space for $10m_{10}^*$ is 0.090 and for $2m_2^*$ is 0.096, compared to 0.110 for the symmetric case m_1^* . The opposite pattern is observed with a < 1, where m_a^* values are similar to, and slightly lower than, m_1^* values across the gene drive configuration space (Fig 6D–6E compared to Fig 6C), but am_a^* values are much lower (Fig 6H–6I compared to Fig 6C). These results indicate that for a given gene drive configuration (*s*, *c*, *h*), maintaining DTEs and avoiding spillovers in a two-deme system with asymmetric migration in which the higher of the two rates is *m*, is comparable to, but slightly more difficult than, in the symmetric case with migration rate *m*. This does not depend on which migration rate is larger.

The maximal impact of the DTE on the non-target population $(q_{2,a}^*)$ follows the critical thresholds in terms of migration from the target to the non-target, am_a^* , with substantially lower impact for a < 1 and slightly lower impact for a > 1, compared to the symmetric case (Fig 6J–6N). Therefore, reduced migration from the target population to the non-target

population (lower *a* values) would result in substantially reduced impact of the gene drive on the non-target population at the DTE.

Gene drive design for applications in nature

Although gene-drive applications are not yet at the stage of deployment in natural settings, it is useful to understand spillover dynamics in specific applications that are already at the design and experimentation stages. Obtaining information on some of the parameters in our models, such as *s* or *m*, for specific systems in natural settings is challenging. However, using available information and making simplifying assumptions to demonstrate the application of the framework in different settings can give insight into the type of considerations that should be taken into account when designing system-specific gene drives. Here, we use two examples to illustrate how the models might behave in the context of scenarios of interest.

Malaria vectors. One of the most discussed potential applications of gene drives is for the control of malaria vectors—*Anopheles* mosquitoes [41–43]. Before the deployment of gene drives in malaria-affected regions, field trials in natural but controlled settings will likely occur [44, 45]. These field trials could, for example, be conducted on islands near the coast of sub-Saharan Africa, in order to simulate conditions similar to those found on the African mainland but in a contained setting [45].

To demonstrate how our theory can be applied, we consider such a field trial, on an island where *Anopheles* mosquitoes occur naturally. Biosafety measures for such an experiment should account for the possibility that mosquitoes carrying the gene-drive allele would find their way to the mainland [45], either through migration processes similar to those that resulted in the initial colonization of the island [46, 47], or through human-assisted migration (e.g., hitchhiking on human transportation).

For *Anopheles* mosquitoes, high gene-drive conversion rates have been reported, approaching full conversion, c = 1 [48]. We adopt the same conversion rate, c = 1, for our hypothetical field trial example. To achieve DTEs, we must ensure that the gene-drive configuration is within B_2 .

For c = 1, irrespective of h, the gene-drive configuration is in B_2 when s > 0.5 (S3 Fig, red regions for c = 1). For s < 0.5, the gene-drive configuration is in A_1 (S3 Fig, green regions for c = 1), and the gene drive is predicted to reach fixation in both populations at any migration rate, perhaps even if only a single individual migrates to the mainland. Therefore, to prevent spillovers and substantial impacts of the gene drive outside the target island, it is crucial to ensure that the fitness of homozygotes of the gene-drive allele, 1 - s, is at most half that of the wild types. This choice ensures that the gene-drive configuration is in B_2 .

Because it is likely that the migration rate is not accurately known, increasing the critical migration threshold increases the likelihood that DTEs can be maintained under unknown migration rates. For full conversion, c = 1, our results indicate that the maximal critical migration threshold attainable is $m^* \approx 0.1$, attained for some intermediate fitness cost s^* (Fig 3A–3C, yellow bands for c = 1). With this configuration, it is possible for the gene drive to be maintained at low frequency on the mainland, even if connectivity between the island and the mainland is as high as $m \approx 0.1$. For migration rates above this critical migration threshold, DTEs no longer exist. Given uncertainty in migration rates, it is therefore prudent to design the gene drive with a fitness cost close to s^* . Our model can provide an initial estimate of s^* , but this parameter would likely need to be ascertained from lab experiments to account for the many biological aspects we do not model here.

Our results for c = 1 indicate that when m^* is exceeded, the gene drive is globally driven to fixation for $s < s^*$, whereas for $s > s^*$ it is globally driven to loss (Fig 4). It is, therefore,

advisable to ensure that *s* is slightly larger than *s*^{*}, as an additional measure to prevent spillover to the mainland. Hence, if migration exceeds the migration limit (e.g. Appendix F in <u>S1 Text</u> and <u>S14 Fig</u>) because of underestimation of the actual migration rates or changes in those rates, the gene drive will fail to spread in both the island and the mainland, rather than spreading in both.

For the configuration c = 1 and $s = s^*$, initiating the gene drive so it converges to the DTE requires a high initial gene-drive frequency on the island, particularly if connectivity to the mainland is substantial; for example, in S12 Fig, the intersection of the border between the yellow and blue regions with the *x*-axis is at $q_1 = 0.63$ for m = 0, and this intersection point increases for higher migration rates. Attaining such high initial frequencies would require releasing many genetically modified individuals, but could perhaps be feasible for a field trial on an island.

Invasive rodents. As a second example, we consider the deployment of gene drives for control of invasive rodents on islands [15, 43]. Here, the consequences of spillovers are critical not only for field trials, but also for the intended purpose of the gene drive, because our final aim is to control the species only on specific islands, without influencing the species in its native range [20, 21, 49]. CRISPR-based gene drives have been shown to work in mice, with lower conversion effectiveness than that reported in invertebrates [40]. For this example, we assume that a gene drive has been developed for the relevant rodent species with c = 0.72, the highest conversion rate reported in mice to date [40].

In this case, unlike for the malaria-vector example with c = 1, we must also consider the dominance parameter *h*. For a recessive gene drive (h = 0), DTEs are possible for gene-drive configurations with 0.5 < s < 0.73 (S3 Fig, red region for c = 0.72 and h = 0). Within this range of s, the DTEs can be maintained under the highest migration rate of $m^* = 0.02$ (Fig 3A, pale yellow band for c = 0.72 around s = 0.6). This critical migration threshold is substantially lower than could be attained in the mosquito example above, meaning that under ideal gene drive configurations, differential targeting can be maintained under higher connectivity in mosquitoes than in rodents. For an additive gene drive (h = 0.5), the fitness-cost range for DTEs is wider, 0.45 < s < 0.84 (S3 Fig, red region for c = 0.72 and h = 0.5), and DTEs can be maintained up to a higher critical threshold of $m^* = 0.06$ (Fig 3B, pale yellow band for c = 0.72). For a dominant gene drive (h = 1), the fitness-cost range is even wider, 0.42 < s < 1 (S3 Fig, red region for c = 0.72 and h = 1), and maximal critical thresholds are even higher, $m^* = 0.08$ (Fig 3C, pale yellow band for c = 0.72). Therefore, unlike in the mosquito example, for rodents we see that *h* plays an important role in determining the parameter ranges in which DTEs can be attained, as well as the critical migration thresholds for DTEs. In this particular example, the critical migration limit is more than three times higher for a dominant gene drive than for a recessive one. Notably, current mouse gene drives are more efficient when conversion occurs in the germline rather than in the zygote [40]. Our analysis of gene drives with germline conversion (Appendix C in S1 Text) indicates that h > 0.5 is required for DTEs to exist, reinforcing the conclusion that dominance of the gene drive allele will likely be crucial for differential targeting in rodents.

For any analysis of gene-drive spillovers, it is important to identify the relevant scale for which the migration rates *m*, defined as migration probabilities of individuals, apply. These rates refer to populations in the region from which migration occurs. In our example, considering a target island with an invasive rodent species and a non-target island with native rodents, and assuming that the rodents migrate mostly via a major entry point on these islands, such as a main port, the subpopulation of the port and its immediate surroundings, and not the population of the entire island, would likely be the appropriate unit for estimating *m*. Migration rates among subpopulations on an island typically exceed migration rates between

populations on different islands. Therefore, for migration between two highly-populated islands, for example, if we measure *m* using the entire population of each island, and not among appropriate subpopulations, then we might significantly underestimate the appropriate *m* for the application of our models.

Discussion

We have explored the possibility of effectively applying a gene drive in a target population while limiting the exposure of a non-target population to spillovers. By investigating equilibria of the evolutionary dynamics, we have shown that with some gene-drive configurations (*s*, *c*, *h*) for which a polymorphic unstable equilibrium exists in the one-deme model (B_2 configurations), it is possible to initiate the gene drive such that differential targeting—higher equilibrium gene-drive allele frequencies in the target population than in the non-target population —is possible. However, we have also shown that for these configurations, upon increasing the migration rate *m*, a sharp transition occurs to a state in which the DTE no longer exists, and the two populations face similar fates, either global loss or global fixation of the gene-drive allele. We also showed that critical migration thresholds are highest for a full conversion gene drive (*c* = 1) with some intermediate fitness cost *s*^{*}.

For most B_2 configurations, other than in a narrow region of the parameter space for (*s*, *c*, *h*), differential targeting can be achieved only for low migration rates (blue regions in Fig 3A–3C). This observation was seen in two-deme models with different assumptions regarding migration asymmetry and timing of migration, selection, and conversion (Appendices B, C, and E in S1 Text). Consequently, for relatively high migration rates, it is unlikely that gene drives could be configured with enough accuracy to provide sufficient confidence that differential targeting could be achieved. In general, our analysis suggests that gene-drive spillovers to the non-target population should therefore be considered likely, and prevention of spillovers should in most cases not rely on differential targeting. Only when the gene-drive configuration (*s*, *c*, *h*) and ecological circumstances have been validated with high accuracy, or when low migration rates between populations can be assured, will it be possible to view differential targeting as a practical measure for limiting spillovers.

We found that the configuration that maximizes the critical migration threshold m^* is also the one that maximizes the impact on the non-target population in the sense of producing the largest q_2^* at equilibrium (Fig 3D–3F). Allowing the selection coefficient to vary, this same configuration also differentiates between scenarios for which increasing migration above m^* results in global loss or global fixation of the gene-drive allele (Fig 4). In practical settings, such breaches of the critical migration threshold may occur as a result of unexpected increases in migration, or from errors in approximating actual migration rates. As a precaution, it would be wise to design gene drives (s, c, h) for which exceeding m^* results in global loss rather than global fixation, in order to avoid full exposure of the non-target population to the gene drive.

Many mathematical gene-drive modeling efforts so far have followed classic populationgenetic models in which allele frequencies do not directly effect the population size (but see [50]). While this assumption is appropriate for most loci of interest in population genetics, gene drives are designed specifically to violate this assumption on very short time-scales. For example, our models assume fixed migration rates, but if a DTE is attained and the gene drive reaches high frequencies in the target population, population-size decline in the target population would likely result in decreased migration from target to non-target populations. Although our models do not directly incorporate the effect of gene-drive frequencies on population dynamics, we can use our investigation of asymmetric migration to provide some insights. Under differential targeting, we expect reduced outgoing migration from the target population but not from the non-target population, because only the target population is substantially affected by the gene drive on a population level. In terms of our model, this reduced migration from the target to non-target populations would result in a decrease in the migration ratio *a* from its initial value. Therefore, with the reservation that we solve for equilibrium conditions and not for dynamic decrease in migration, the observation that for given gene-drive configurations, critical thresholds are not breached as *a* decreases (Fig 6A–6I) implies that differential targeting may persist through the formation of asymmetric migration. In addition, we observe that as *a* decreases, the maximal impact on the non-target population $(q_{2,a}^*)$ decreases (Fig 6J–6N). Therefore, differential targeting may be a relevant strategy even when considering the demographic impact of gene drives on the target population.

For the two potential applications of gene-drive deployment, malaria-vector field trials and invasive rodents on islands, we observed that the conversion efficiency in the organism affects the design configurations for limiting spillovers. For very high conversion rates, as have been attained in mosquitoes, the main consideration is the selection coefficient of the gene drive, and migration limits for maintaining differential targeting can be relatively high. However, in the case of rodents, where conversion efficiency is currently lower, we see that the degree of dominance of the gene-drive allele has a strong effect on the migration limits. The maximal migration limits are lower than for the mosquito example, with higher dominance of the gene-drive allele resulting in higher migration limits. This result suggests that, with minimizing the risk of spillover as a goal, dominance should be given a priority in genedrive design for rodents, while it is not as important for malaria vectors, particularly for gene-drive conversion in the germline. These examples also highlight that the gene-drive selection coefficient should be chosen with the other parameters, particularly conversion rates, in mind.

Before CRISPR-based gene drives can be efficiently and safely applied, many challenges, such as the potential evolution of resistance to the gene-drive modification [51-57], must be overcome. For the type of differential targeting described here, an accurate configuration of gene-drive parameters, *s*, *c*, and *h*, is required. However, it is unclear whether such accuracy is feasible [10, 40, 48, 58-61]. Another difficulty for differential targeting is that implementation might require a concentrated effort to initiate gene drive deployment in the basin of attraction of the desired equilibrium (the DTE). For example, in the malaria-vector field trial case examined here, the gene-drive allele frequency must be pushed above 0.6 before the system is expected to converge to the DTE. This effort would therefore involve engineering and releasing a large and potentially unfeasible number of engineered individuals. A third problem is that we do not yet understand the consequences of more complex population structure, involving more than two populations or populations that are internally structured. Given these difficulties, to understand and mitigate the dangers of spillovers, it will be important to study more elaborate population structures in models of gene drives.

Several approaches have been recently proposed for mitigating spillovers, involving complex gene drive architectures and deployment strategies [39, 62–69], as well as for implementing countermeasures to halt an ongoing gene drive [10, 70–72]. A few of these genedrive architectures have been demonstrated in laboratory settings [10, 38, 53, 60, 73]. However, performance of mitigation strategies in local confinement of a gene drive has only been examined through proof-of-concept mathematical models, and their behaviors outside the specific population and environmental conditions examined are still not well understood. We have investigated the prospects of differential targeting as a confinement strategy under a standard homing CRISPR-based gene drive construct, but our model could potentially be extended to incorporate more complex constructs, and be deployed under various strategies. Therefore, given the dangers and many unknowns of gene-drive dynamics, we suggest that multiple safeguards should be used in parallel. In this way, differential targeting, sophisticated genetic architectures, informed deployment strategies, and countermeasures for halting the spread of already-active gene drives could be used in concert to reduce genedrive spillover risks.

Mathematical investigations of the type conducted here are intended to reveal qualitative behaviors of expected dynamics, and can provide directions for mitigating gene drive spillovers by configuring gene-drive parameters. However, the many simplifying assumptions of these models mean that any conclusions, particularly those involving specific parameter values, need to be thoroughly explored in more realistic models, in lab experiments, and ideally in natural conditions, before being considered viable options for gene-drive deployment. In addition, we have studied equilibrium conditions to understand the direction to which twodeme systems would tend, but we did not study explicit transient dynamics, which are important for understanding, for example, how long it would take for the system to converge to equilibria. Therefore, to improve our understanding of gene drive spillovers, further exploration of gene-drive models incorporating various gene-drive constructs, deployment strategies, and ecological features, as well as lab experimentation with structured caged populations, is needed.

The existence of scenarios permitting differential targeting should not be read as a call for applications of gene drives in wild settings, nor as a blueprint for CRISPR-based gene-drive design. Instead, we believe the consequences of gene-drive application and the potential for spillovers should be further discussed in the scientific community and in society at large before actual gene-drive applications are realized, and a goal of this paper is to facilitate and inform these discussions.

Methods

Equilibria

Equilibria were computed by numerically solving the systems of equations in Eq.3, with the equilibrium conditions $q'_1 = q_1$ and $q'_2 = q_2$, using Mathematica software [74].

Stability of equilibria

For each equilibrium (\hat{q}_1, \hat{q}_2) , its stability was determined by examining how the trajectories of the allele frequencies in the two populations behave in its vicinity in allele frequency space.

The system of discrete-time recursion equations that describe the change of allele frequencies over a single generation are:

$$q'_{1} = f_{1}(q_{1}, q_{2})$$

$$q'_{2} = f_{2}(q_{1}, q_{2}),$$
(5)

where f_i (for i = 1, 2) is defined in Eq.3 or Equation S2 in S1 Text. To analyze the stability of the system in the vicinity of an equilibrium (\hat{q}_1, \hat{q}_2) , we linearized the nonlinear functions f_1 and f_2 by taking the first-order approximation of the Taylor expansion about the equilibrium. From this linearization, in order to understand the consequences of small perturbations from the equilibrium, we examined the matrix of the partial derivatives of f_1 and f_2 , evaluated at the equilibrium (i.e., the Jacobian matrix of Eq 5):

$$\mathbf{J} = \begin{bmatrix} \frac{\partial f_1}{\partial q_1} & \frac{\partial f_1}{\partial q_2} \\ \frac{\partial f_2}{\partial q_1} & \frac{\partial f_2}{\partial q_2} \end{bmatrix}.$$
 (6)

The local stability of the equilibrium point (\hat{q}_1, \hat{q}_2) is determined by the two eigenvalues λ_1 and λ_2 of **J** evaluated at (\hat{q}_1, \hat{q}_2) . If both eigenvalues have modulus less than 1 ($|\lambda_1| < 1$ and $|\lambda_2| < 1$), then the equilibrium point (\hat{q}_1, \hat{q}_2) is stable. If at least one eigenvalue has modulus greater than 1, then the equilibrium point is unstable [75]. For the special case with $|\lambda_1| = 1$ or $|\lambda_2| = 1$, see [75].

To generate Fig 2 and S4–S9 Figs, we computed the equilibria and determined their stability for all (*s*, *c*, *h*) combinations, with *s* and *c* ranging from 0 to 1 in increments of 0.01, and h = 0, 0.5 or 1. To generate Figs 3 and 4, these configurations were examined for each migration rate *m* between 0 and 0.15 at increments of 0.0001. Basins of attraction in Fig 2 and S14 Fig were computed by iterating the recursive equations with different initial conditions to identify the equilibrium to which they converge.

For a stable equilibrium, there exists a basin of attraction in allele frequency space around the equilibrium [76], such that any trajectory starting within the basin eventually converges to the stable equilibrium point. To determine basins of attraction for a configuration (*s*, *c*, *h*) and migration rate *m*, we started from a grid of points in frequency space, and for each point (*q*₁, *q*₂), we tracked its trajectory for 1000 generations by iterating Eq 3. We denote the endpoint of the trajectory with initial condition (*q*₁, *q*₂) by (*q*₁, *q*₂)₁₀₀₀. The attractor associated with (*q*₁, *q*₂) was defined as the equilibrium (\hat{q}_1, \hat{q}_2) that was at a Euclidean distance less than 0.001 in frequency space from (*q*₁, *q*₂)₁₀₀₀. If there were no such equilibria or more than one such equilibrium, then the attractor of (*q*₁, *q*₂) was left undefined. The basins of attraction were computed for a grid of (*q*₁, *q*₂) points at a resolution of 0.01 × 0.01 in frequency space. For all computations used to generate Figs 2, 4, and 5, attractors were defined for all points computed.

Supporting information

S1 Text. Section A explains the evolutionary dynamics of a gene drive in a one-deme model. Section B contains a mathematical model and numerical results in the two-deme model with selection before migration. Section C contains a mathematical model and results in the twodeme model with conversion after selection and migration. Section D details the mathematical modeling and simulation procedures for adding genetic drift and computing the safety radius. Section E presents the results for recessive and dominant gene drives with asymmetric migration. Section F presents an analysis of the malaria-vector example. (PDF)

S1 Fig. Schematic depiction of a two-deme gene-drive model with selection occurring before migration. For interpretation of arrow colors, see <u>Fig 1</u> in the main text. (PNG)

S2 Fig. Critical migration thresholds for differential targeting in the two-deme model with selection before migration. (A–C) Maximal migration rates m^* for which a differential-targeting equilibrium (DTE) exists. The colored regions denote the configurations for which a DTE exists (B_2 configurations), and the regions in white denote configurations for which

differential targeting of the demes is not possible (A_1 , A_2 , and B_1). Over most of the parameter space, a DTE exists only for low migration rates (in blue), and only in a narrow band do DTEs exist with migration rates above $m^* = 0.05$ (light yellow). (D–F) Maximal gene-drive frequencies in the non-target population, q_2^* , at differential-targeting equilibria (DTEs). q_2^* values are correlated with m^* values. The figure design follows Fig 3. (PNG)

S3 Fig. Subsets of gene-drive configurations, according to results derived from the onedeme model (Eq 1). Green: A_1 configurations, with two trivial equilibria, $\hat{q} = 0$ (loss) unstable and $\hat{q} = 1$ (fixation) stable; yellow: A_2 configurations, with two trivial equilibria, $\hat{q} = 0$ stable and $\hat{q} = 1$ unstable; blue: B_1 configurations, with 1 stable non-trivial equilibrium and 2 unstable trivial equilibria; red: B_2 configurations, with 1 unstable non-trivial equilibrium and 2 stable trivial equilibria. The panels show results for 11 values of *h* and all possible values of *s* and *c*. (PNG)

S4 Fig. Partitioning of the parameter space by the characterization of the equilibria for the two-deme migration-before-selection model (Eq 3), for recessive gene drives (h = 0), under different migration rates. Green—9 equilibria, 4 stable and 5 unstable, one of which is a DTE (stable and $\hat{q}_1 > \hat{q}_2$); purple—9 equilibria, 8 unstable and 1 stable and symmetric ($\hat{q}_1 = \hat{q}_2$); blue—5 equilibria, 2 stable trivial (global fixation and global loss) and 3 unstable; red—3 equilibria, 2 stable trivial and 1 unstable; orange—3 equilibria, 2 unstable trivial and 1 stable symmetric; white—2 trivial equilibria, 1 stable and 1 unstable (A_1 and A_2 configurations). In the bottom white region, the stable equilibrium is global fixation of the gene-drive allele (A_1 configuration), and in the top white region, the stable equilibrium is global loss (A_2 configuration). The only gene-drive configurations for which differential targeting is possible appear in green. (PNG)

S5 Fig. Partitioning of the parameter space by the characterization of the equilibria for the two-deme migration-before-selection model (Eq 3), for additive gene drives (h = 0.5), under different migration rates. Green—9 equilibria, 4 stable and 5 unstable, one of which is a DTE (stable and $\hat{q}_1 > \hat{q}_2$); purple—9 equilibria, 8 unstable and 1 stable and symmetric ($\hat{q}_1 = \hat{q}_2$); blue—5 equilibria, 2 stable trivial (global fixation and global loss) and 3 unstable; red—3 equilibria, 2 stable trivial and 1 unstable; white—2 trivial equilibria, 1 stable and 1 unstable (A_1 and A_2 configurations). In the bottom white region, the stable equilibrium is global fixation of the gene-drive allele (A_1 configuration), and in the top white region, the stable equilibrium is global loss (A_2 configuration). The only gene-drive configurations for which differential targeting is possible appear in green. (PNG)

S6 Fig. Partitioning of the parameter space by the characterization of the equilibria for the two-deme migration-before-selection model (Eq 3), for dominant gene drives (h = 1), under different migration rates. Green—9 equilibria, 4 stable and 5 unstable, one of which is a DTE (stable and $\hat{q}_1 > \hat{q}_2$); purple—9 equilibria, 8 unstable and 1 stable and symmetric ($\hat{q}_1 = \hat{q}_2$); blue—5 equilibria, 2 stable trivial (global fixation and global loss) and 3 unstable; red—3 equilibria, 2 stable trivial and 1 unstable; white— A_1 configuration, with 1 stable (global fixation) and 1 unstable (global loss) equilibrium. The only gene-drive configurations for which differential targeting is possible appear in green. (PNG)

S7 Fig. Partitioning of the parameter space by the characterization of the equilibria for the two-deme selection-before-migration model (Eq S2), for recessive gene drives (h = 0),

under different migration rates. Green—9 equilibria, 4 stable and 5 unstable, one of which is a DTE (stable and $\hat{q}_1 > \hat{q}_2$); purple—9 equilibria, 8 unstable and 1 stable and symmetric $(\hat{q}_1 = \hat{q}_2)$; blue—5 equilibria, 2 stable trivial (global fixation and global loss) and 3 unstable; red—3 equilibria, 2 stable trivial and 1 unstable; orange—3 equilibria, 2 unstable trivial and 1 stable symmetric; white—2 trivial equilibria, 1 stable and 1 unstable (A_1 and A_2 configurations). In the bottom white region, the stable equilibrium is global fixation of the gene-drive allele (A_1 configuration), and in the top white region, the stable equilibrium is global loss (A_2 configuration). The only gene-drive configurations for which differential targeting is possible appear in green.



S8 Fig. Partitioning of the parameter space by the characterization of the equilibria for the two-deme selection-before-migration model (Eq S2), for additive gene drives (h = 0.5), under different migration rates. Green—9 equilibria, 4 stable and 5 unstable, one of which is a DTE (stable and $\hat{q}_1 > \hat{q}_2$); purple—9 equilibria, 8 unstable and 1 stable and symmetric ($\hat{q}_1 = \hat{q}_2$); blue—5 equilibria, 2 stable trivial (global fixation and global loss) and 3 unstable; red—3 equilibria, 2 stable trivial and 1 unstable; white—2 trivial equilibria, 1 stable and 1 unstable (A_1 and A_2 configurations). In the bottom white region, the stable equilibrium is global fixation of the gene-drive allele (A_1 configuration), and in the top white region, the stable equilibrium is global loss (A_2 configuration). The only gene-drive configurations for which differential targeting is possible appear in green. (PNG)

S9 Fig. Partitioning of the parameter space by the characterization of the equilibria for the two-deme selection-before-migration model (Eq S2), for dominant gene drives (h = 1), under different migration rates. Green—9 equilibria, 4 stable and 5 unstable, one of which is a DTE (stable and $\hat{q}_1 > \hat{q}_2$); purple—9 equilibria, 8 unstable and 1 stable and symmetric ($\hat{q}_1 = \hat{q}_2$); blue—5 equilibria, 2 stable trivial (global fixation and global loss) and 3 unstable; red—3 equilibria, 2 stable trivial and 1 unstable; white— A_1 configuration, with 1 stable (global fixation) and 1 unstable (global loss) equilibrium. The only gene-drive configurations for which differential targeting is possible appear in green. (PNG)

S10 Fig. Perturbation from DTEs for the model with selection before migration (Eq S3). Results shown for gene drives with full conversion, c = 1, and for different migration rates m and selection coefficients s. White regions denote scenarios for which $m > m^*$. (A) The probability of escape from the DTE due to genetic drift, defined as the probability of departing from the attraction basin of the DTE over 100 generations with genetic drift, in a Wright-Fisher population with $N_e = 100$. The black line denotes 5% probability of escape. Probabilities were estimated from 1000 simulated replicates. (B) The safety radius of the DTE for the m-before-s model. (PNG)

S11 Fig. Impact of genetic drift on the probability of escape. The critical migration threshold m^* for the two-deme model in Eq 3 with c = 1 is shown in red. Other curves show the threshold for probability of escape >0.05% for different effective population sizes of the demes, for the same model, computed with 1000 simulated iterations of Eq S3 for each gene-drive configuration (equivalent to black line in Fig 5A and S10A Fig). $N_e = 200$ in blue; $N_e = 50$ in yellow; $N_e = 10$ in purple. Critical migration thresholds are effectively lower with lower effective population sizes.

(PNG)

S12 Fig. Critical migration thresholds m_a^* and impact on non-target population $q_{2,a}^*$ with asymmetric migration for recessive gene drives (h = 0). (A–E) Critical migration thresholds from the non-target deme to the target deme, m_a^* , for migration ratio a. (F–I) Critical migration thresholds expressed as migration from target to non-target demes, am_a^* , for migration ratio a. (J–N) Maximal gene-drive frequencies in the non-target population at the DTE, $q_{2,a}^*$, with migration ratio a. (PNG)

S13 Fig. Critical migration thresholds m_a^* and impact on non-target population $q_{2,a}^*$ with asymmetric migration for dominant gene drives (h = 1). (A–E) Critical migration thresholds from the non-target deme to the target deme, m_a^* , for migration ratio a. (F–I) Critical migration thresholds expressed as migration from target to non-target demes, am_a^* , for migration ratio a. (J–N) Maximal gene-drive frequencies in the non-target population at the at DTE, $q_{2,a}^*$, with migration ratio a.

(PNG)

S14 Fig. Equilibria and basins of attractions for the malaria-vector example. Shown are results for the *m*-before-*s* model with a gene-drive configuration (*s*, *c*, *h*) of c = 1, s = 0.73 and arbitrary *h*, as per the malaria-vector example in the main text. The circles show the equilibria. The colored regions show the attraction basins, with the basin colors corresponding to the stable equilibria. The arrows show the vector field that describes the magnitude and direction of the change in allele frequencies at each point in frequency space. The differential-targeting equilibrium (DTE) is the stable yellow equilibrium point, which exists for migration rates lower than $m^* \approx 0.093$. (PNG)

Author Contributions

Conceptualization: Gili Greenbaum, Jaehee Kim.

Data curation: Gili Greenbaum.

Formal analysis: Gili Greenbaum, Jaehee Kim.

Funding acquisition: Marcus W. Feldman, Noah A. Rosenberg.

Investigation: Gili Greenbaum, Jaehee Kim.

Methodology: Gili Greenbaum, Jaehee Kim.

Software: Gili Greenbaum.

Supervision: Marcus W. Feldman, Noah A. Rosenberg.

Validation: Gili Greenbaum, Marcus W. Feldman, Noah A. Rosenberg, Jaehee Kim.

Visualization: Gili Greenbaum, Jaehee Kim.

Writing - original draft: Gili Greenbaum, Jaehee Kim.

Writing – review & editing: Gili Greenbaum, Marcus W. Feldman, Noah A. Rosenberg, Jaehee Kim.

References

 Hartl DL. Analysis of a general population genetic model of meiotic drive. Evolution. 1970; 24(3):538– 545. https://doi.org/10.1111/j.1558-5646.1970.tb01789.x

- Thomson GJ, Feldman MW. Population genetics of modifiers of meiotic drive. II. Linkage modification in the segregation distortion system. Theoretical Population Biology. 1974; 5(2):155–162. <u>https://doi.org/ 10.1016/0040-5809(74)90038-0</u>
- Thomson GJ, Feldman MW. Population genetics of modifiers of meiotic drive: IV. On the evolution of sex-ratio distortion. Theoretical Population Biology. 1975; 8(2):202–211. <u>https://doi.org/10.1016/0040-5809(75)90032-5</u>
- Hartl DL. Modifier theory and meiotic drive. Theoretical Population Biology. 1975; 7(2):168–174. https://doi.org/10.1016/0040-5809(75)90012-X
- Thomson GJ, Feldman MW. Population genetics of modifiers of meiotic drive III. Equilibrium analysis of a general model for the genetic control of segregation distortion. Theoretical Population Biology. 1976; 10(1):10–25. https://doi.org/10.1016/0040-5809(76)90003-4
- Charlesworth B, Hartl DL. Population dynamics of the segregation distorter polymorphism of *Drosophila* melanogaster. Genetics. 1978; 89(1):171–92.
- Deredec A, Burt A, Godfray HCJ. The population genetics of using homing endonuclease genes in vector and pest management. Genetics. 2008; 179(4):2013–2026. https://doi.org/10.1534/genetics.108. 089037
- 8. Marshall JM, Hay BA. Confinement of gene drive systems to local populations: A comparative analysis. Journal of Theoretical Biology. 2012; 294:153–171. https://doi.org/10.1016/j.jtbi.2011.10.032
- Unckless RL, Messer PW, Connallon T, Clark AG. Modeling the manipulation of natural populations by the mutagenic chain reaction. Genetics. 2015; 201(2):425–431. <u>https://doi.org/10.1534/genetics.115.</u> 177592
- Gantz VM, Bier E. The mutagenic chain reaction: A method for converting heterozygous to homozygous mutations. Science. 2015; 348(6233):442–444. https://doi.org/10.1126/science.aaa5945
- Oye KA, Esvelt K, Appleton E, Catteruccia F, Church G, Kuiken T, et al. Regulating gene drives. Science. 2014; 345(6197):626–628. https://doi.org/10.1126/science.1254287 PMID: 25035410
- Champer J, Buchman A, Akbari OS. Cheating evolution: Engineering gene drives to manipulate the fate of wild populations. Nature Reviews Genetics. 2016; 17(3):146–159. https://doi.org/10.1038/nrg.2015. 34
- Piaggio AJ, Segelbacher G, Seddon PJ, Alphey L, Bennett EL, Carlson RH, et al. Is it time for synthetic biodiversity conservation? Trends in Ecology and Evolution. 2017; 32(2):97–107. <u>https://doi.org/10.1016/j.tree.2016.10.016 PMID: 27871673</u>
- Prowse TAA, Cassey P, Ross JV, Pfitzner C, Wittmann TA, Thomas P. Dodging silver bullets: Good CRISPR gene-drive design is critical for eradicating exotic vertebrates. Proceedings of the Royal Society B: Biological Sciences. 2017; 284(1860):20182048.
- Leitschuh CM, Kanavy D, Backus GA, Valdez RX, Serr M, Pitts EA, et al. Developing gene drive technologies to eradicate invasive rodents from islands. Journal of Responsible Innovation. 2018; 5:S121– S138. https://doi.org/10.1080/23299460.2017.1365232
- Webber BL, Raghu S, Edwards OR. Is CRISPR-based gene drive a biocontrol silver bullet or global conservation threat? Proceedings of the National Academy of Sciences. 2015; 112(34):10565–10567.
- 17. Esvelt KM, Gemmell NJ. Conservation demands safe gene drive. PLOS Biology. 2017; 15(11):1–8.
- Noble C, Adlam B, Church GM, Esvelt KM, Nowak MA. Current CRISPR gene drive systems are likely to be highly invasive in wild populations. eLife. 2018; 7:e33423. https://doi.org/10.7554/eLife.33423
- Rode NO, Estoup A, Bourguet D, Courtier-Orgogozo V, Débarre F. Population management using gene drive: molecular design, models of spread dynamics and assessment of ecological risks. Conservation Genetics. 2019; 20:671–690. https://doi.org/10.1007/s10592-019-01165-5
- 20. Owens B. The big cull: Can New Zealand pull off an audacious plan to get rid of invasive predators by 2050? Nature. 2017; 541:148–150.
- Dearden PK, Gemmell NJ, Mercier OR, Lester PJ, Scott MJ, Newcomb RD, et al. The potential for the use of gene drives for pest control in New Zealand: a perspective. Journal of the Royal Society of New Zealand. 2018; 48(4):225–244. https://doi.org/10.1080/03036758.2017.1385030
- Sandler L, Novitski E. Meiotic drive as an evolutionary force. The American Naturalist. 1957; 91 (857):105–110. https://doi.org/10.1086/281969
- Lyttle TW. Cheaters sometimes prosper: distortion of mendelian segregation by meiotic drive. Trends in Genetics. 1993; 9(6):205–210. https://doi.org/10.1016/0168-9525(93)90120-7
- 24. Lewontin RC, Dunn LC. The evolutionary dynamics of a polymorphism in the house mouse. Genetics. 1960; 45(6):705. https://doi.org/10.1093/genetics/45.6.705
- Lewontin RC. Interdeme selection controlling a polymorphism in the house mouse. The American Naturalist. 1962; 96(887):65–78. https://doi.org/10.1086/282208

- Karlin S, McGregor J. Equilibria for genetic systems with weak interaction. In: Proceedings of the Sixth Berkeley Symposium on Mathematical Statistics and Probability, Volume 4: Biology and Health. Berkeley, CA; 1972. p. 79–87.
- Prout T, Bundgaard J, Bryant S. Population genetics of modifiers of meiotic drive. I. The solution of a special case and some general implications. Theoretical Population Biology. 1973; 4(4):446–465. https://doi.org/10.1016/0040-5809(73)90020-8
- Curtsinger JW, Feldman MW. Experimental and theoretical analysis of the 'sex-ratio' polymorphism in Drosophila pseudoobscura. Genetics. 1980; 94(2):445–466.
- 29. Hiraizumi Y, Sandler L, Crow JF. Meiotic drive in natural populations of *Drosophila melanogaster*. III. Populational implications of the segregation-distorter locus. Evolution. 1960; 14(4):433–444. <u>https://doi.org/10.1111/j.1558-5646.1960.tb03111.x</u>
- Davis S, Bax N, Grewe P. Engineered underdominance allows efficient and economical introgression of traits into pest populations. Journal of Theoretical Biology. 2001; 212(1):83–98. <u>https://doi.org/10.1006/</u> jtbi.2001.2357
- Magori K, Gould F. Genetically engineered underdominance for manipulation of pest populations: A deterministic model. Genetics. 2006; 172(4):2613–2620. <u>https://doi.org/10.1534/genetics.105</u>. 051789
- Altrock PM, Traulsen A, Reeves RG, Reed FA. Using underdominance to bi-stably transform local populations. Journal of Theoretical Biology. 2010; 267(1):62–75. https://doi.org/10.1016/j.jtbi.2010.08.004
- **33.** Tanaka H, Stone HA, Nelson DR. Spatial gene drives and pushed genetic waves. Proceedings of the National Academy of Science. 2017; 114(32):8752–8457.
- Karlin S, McGregor J, Phelps MP, Seeb LW, Seeb JE, Rode NO, et al. Application of method of small parameters to multi-niche population genetic models. Theoretical Population Biology. 1972; 3(2):186– 209. https://doi.org/10.1016/0040-5809(72)90026-3 PMID: 4667083
- Karlin S, McGregor J. Polymorphisms for genetic and ecological systems with weak coupling. Theoretical Population Biology. 1972; 3(2):210–238. https://doi.org/10.1016/0040-5809(72)90027-5
- Huang Y, Lloyd AL, Legros M, Gould F. Gene-drive into insect populations with age and spatial structure: A theoretical assessment. Evolutionary Applications. 2011; 4(3):415–428. <u>https://doi.org/10.1111/j.1752-4571.2010.00153.x</u>
- Marshall JM, Hay BA. Inverse Medea as a novel gene drive system for local population replacement: A theoretical analysis. Journal of Heredity. 2011; 102(3):336–341. https://doi.org/10.1093/jhered/esr019
- Akbari OS, Matzen KD, Marshall JM, Huang H, Ward CM, Hay BA. A synthetic gene drive system for local, reversible modification and suppression of insect populations. Current Biology. 2013; 23(8):671– 677. https://doi.org/10.1016/j.cub.2013.02.059
- Dhole S, Vella MR, Lloyd AL, Gould F. Invasion and migration of spatially self-limiting gene drives: A comparative analysis. Evolutionary Applications. 2018; 11(5):794–808. https://doi.org/10.1111/eva. 12583
- Grunwald HA, Gantz VM, Poplawski G, Xu XRS, Bier E, Cooper KL. Super-Mendelian inheritance mediated by CRISPR–Cas9 in the female mouse germline. Nature. 2019; 566:105–109. <u>https://doi.org/10. 1038/s41586-019-0875-2</u>
- 41. Alphey L. Can CRISPR-Cas9 gene drives curb malaria? Nature Biotechnology. 2016; 34(2):149–150.
- Burt A, Coulibaly M, Crisanti A, Diabate A, Kayondo JK. Gene drive to reduce malaria transmission in sub-Saharan Africa. Journal of Responsible Innovation. 2018; 5:S66–S80. https://doi.org/10.1080/ 23299460.2017.1419410
- Scudellari M. Self-destructing mosquitoes and sterilized rodents: The promise of gene drives. Nature. 2019; 571(7764):160–162. https://doi.org/10.1038/d41586-019-02087-5
- 44. Kim E, Ilagan JO, Liang Y, Daubner GM, Stanley C, Ramakrishnan A, et al. Rules of the road for insect gene drive research. Nature biotechnology. 2016; 27(5):617–630.
- 45. James S, Collins FH, Welkhoff PA, Emerson C, J Godfray HC, Gottlieb M, et al. Pathway to deployment of gene drive mosquitoes as a potential biocontrol tool for elimination of malaria in sub-Saharan Africa: Recommendations of a scientific working group. American Journal of Tropical Medicine and Hygiene. 2018; 98(Suppl 6):1–49. https://doi.org/10.4269/ajtmh.18-0083
- 46. Pinto J, Donnelly MJ, Sousa CA, Malta-Vacas J, Gil V, Ferreira C, et al. An island within an island: Genetic differentiation of *Anopheles gambiae* in São Tomé, West Africa, and its relevance to malaria vector control. Heredity. 2003; 91(4):407–414. <u>https://doi.org/10.1038/sj.hdy.6800348</u> PMID: 14512957
- Miles A, Harding NJ, Bottà G, Clarkson CS, Antão T, Kozak K, et al. Genetic diversity of the African malaria vector *Anopheles gambiae*. Nature. 2017; 552:96–100. https://doi.org/10.1038/nature24995

- Gantz VM, Jasinskiene N, Tatarenkova O, Fazekas A, Macias VM, Bier E, et al. Highly efficient Cas9mediated gene drive for population modification of the malaria vector mosquito *Anopheles stephensi*. Proceedings of the National Academy of Sciences. 2015; 112(49):E6736–E6743. https://doi.org/10. 1073/pnas.1521077112 PMID: 26598698
- Campbell KJ, Saah JR, Brown PR, Godwin J, Gould F, Howald GR, et al. Island invasives: scaling up to meet the challenge international conference on island invasives. Proceedings of the International Conference on Island Invasives. 2017; p. 6–14.
- Rode NO, Courtier-Orgogozo V, Débarre F. Can a population targeted by a CRISPR-based homing gene drive be rescued? G3: Genes, Genomes, Genetics. 2020; 10(9):3403–3415.
- Unckless RL, Clark AG, Messer PW. Evolution of resistance against CRISPR/Cas9 gene drive. Genetics. 2017; 205(2):827–841. https://doi.org/10.1534/genetics.116.197285
- Church GM, Noble C, Nowak MA, Esvelt KM, Olejarz J. Evolutionary dynamics of CRISPR gene drives. Science Advances. 2017; 3(4):e1601964. https://doi.org/10.1126/sciadv.1601964
- Champer J, Reeves R, Oh SY, Liu C, Liu J, Clark AG, et al. Novel CRISPR/Cas9 gene drive constructs reveal insights into mechanisms of resistance allele formation and drive efficiency in genetically diverse populations. PLOS Genetics. 2017; 13(7):1–18. https://doi.org/10.1371/journal.pgen.1006796 PMID: 28727785
- Gemmell NJ, Tompkins DM. Gene drives and rodent control: Response to Piaggio et al. Trends in Ecology and Evolution. 2017; 32(5):314–315. https://doi.org/10.1016/j.tree.2017.03.005 PMID: 28359574
- Zentner GE, Wade MJ, Drury DW, Siniard DJ, Dapper AL. CRISPR/Cas9 gene drives in genetically variable and nonrandomly mating wild populations. Science Advances. 2017; 3(5):e1601910. <u>https://doi.org/10.1126/sciadv.1601910</u>
- Marshall JM, Buchman A, Sánchez CHM, Akbari OS. Overcoming evolved resistance to populationsuppressing homing-based gene drives. Scientific Reports. 2017; 7:3776. <u>https://doi.org/10.1038/</u> s41598-017-02744-7
- Champer J, Liu J, Oh SY, Reeves R, Luthra A, Oakes N, et al. Reducing resistance allele formation in CRISPR gene drive. Proceedings of the National Academy of Sciences. 2018; 115(21):5522–5527. https://doi.org/10.1073/pnas.1720354115 PMID: 29735716
- Hammond A, Galizi R, Kyrou K, Simoni A, Siniscalchi C, Katsanos D, et al. A CRISPR-Cas9 gene drive system targeting female reproduction in the malaria mosquito vector *Anopheles gambiae*. Nature Biotechnology. 2016; 34(1):78–83. https://doi.org/10.1038/nbt.3439 PMID: 26641531
- 59. Shapiro RS, Chavez A, Porter CBMM, Hamblin M, Kaas CS, DiCarlo JE, et al. A CRISPR-Cas9-based gene drive platform for genetic interaction analysis in *Candida albicans*. Nature Microbiology. 2017; 3 (1):73–82. https://doi.org/10.1038/s41564-017-0043-0 PMID: 29062088
- Buchman A, Marshall JM, Ostrovski D, Yang T, Akbari OS. Synthetically engineered Medea gene drive system in the worldwide crop pest *Drosophila suzukii*. Proceedings of the National Academy of Sciences. 2018; 18(115):4725–4730.
- Kyrou K, Hammond AM, Galizi R, Kranjc N, Burt A, Beaghton AK, et al. A CRISPR-Cas9 gene drive targeting doublesex causes complete population suppression in caged *Anopheles gambiae* mosquitoes. Nature Biotechnology. 2018;(36):1062–1066. https://doi.org/10.1038/nbt.4245 PMID: 30247490
- Dhole S, Lloyd AL, Gould F. Tethered homing gene drives: A new design for spatially restricted population replacement and suppression. Evolutionary Applications. 2019; 12(8):1688–1702. <u>https://doi.org/ 10.1111/eva.12827</u>
- Sudweeks J, Hollingsworth B, Blondel DV, Campbell KJ, Dhole S, Eisemann JD, et al. Locally Fixed Alleles: A method to localize gene drive to island populations. Scientific Reports. 2019; 9(1):15821. https://doi.org/10.1038/s41598-019-51994-0 PMID: 31676762
- Champer J, Lee E, Yang E, Liu C, Clark AG, Messer PW. A toxin-antidote CRISPR gene drive system for regional population modification. Nature Communications. 2020; 11(1):1082. https://doi.org/10. 1038/s41467-020-14960-3
- Akbari OS, Bellen HJ, Bier E, Bullock SL, Burt A, Church GM, et al. Safeguarding gene drive experiments in the laboratory. Science. 2015; 349(6251):927–929. <u>https://doi.org/10.1126/science.aac7932</u> PMID: 26229113
- DiCarlo JE, Chavez A, Dietz SL, Esvelt KM, Church GM. Safeguarding CRISPR-Cas9 gene drives in yeast. Nature Biotechnology. 2015; 33(12):1250–1255. https://doi.org/10.1038/nbt.3412
- Marshall JM, Akbari OS. Can CRISPR-based gene drive be confined in the wild? A question for molecular and population biology. ACS Chemical Biology. 2018; 13(2):424–430. https://doi.org/10.1021/ acschembio.7b00923
- Champer J, Chung J, Lee YL, Liu C, Yang E, Wen Z, et al. Molecular safeguarding of CRISPR gene drive experiments. eLife. 2019; 8:1–10. https://doi.org/10.7554/eLife.41439 PMID: 30666960

- 69. Noble C, Min J, Olejarz J, Buchthal J, Chavez A, Smidler AL, et al. Daisy-chain gene drives for the alteration of local populations. Proceedings of the National Academy of Science. 2019; 116(17):8275–8282. https://doi.org/10.1073/pnas.1716358116 PMID: 30940750
- 70. Wu B, Luo L, Gao XJ. Cas9-triggered chain ablation of *cas9* as a gene drive brake. Nature Biotechnology. 2016; 34(2):137–138. https://doi.org/10.1038/nbt.3444
- Vella MR, Gunning CE, Lloyd AL, Gould F. Evaluating strategies for reversing CRISPR-Cas9 gene drives. Scientific Reports. 2017; 7:11038. https://doi.org/10.1038/s41598-017-10633-2
- 72. Xu XRS, Bulger EA, Gantz VM, Klanseck C, Heimler SR, Auradkar A, et al. Active genetic neutralizing elements for halting or deleting gene drives. Molecular Cell. 2020;In press. <u>https://doi.org/10.1016/j.molcel.2020.09.003</u> PMID: 32949493
- Reeves RG, Bryk J, Altrock PM, Denton JA, Reed FA. First steps towards underdominant genetic transformation of insect populations. PLOS ONE. 2014; 9(5):1–9.
- 74. Wolfram Research I, Wolfram Research. Mathematica, Version 11.3. 2018.
- 75. Day T, Otto SP. A Biologist's Guide to Mathematical Modeling in Ecology and Evolution. Princeton, NJ: Princeton University Press; 2007.
- Elaydi S. Discrete Chaos: With Applications in Science and Engineering. 2nd ed. Chapman and Hall/ CRC; 2007.